Author’s response to reviews

Title: `Whose Shoes? ` Can an educational board game engage Ugandan men in pregnancy and childbirth?

Authors:
Alice Ladur (aladur@bournemouth.ac.uk)
Edwin van Teijlingen (evteijlingen@bournemouth.ac.uk)
Vanora Hundley (vhundley@bournemouth.ac.uk)

Version: 3 Date: 06 Feb 2018

Author’s response to reviews:

Dear Editor,

Thank you for the kind feedback on the manuscript "Whose Shoes? ` Can an educational board game engage Ugandan men in pregnancy and childbirth?"

A point by point response is included in the re-submission.

Yours Faithfully

Alice Ladur